\begin{table}%t2
\par
\caption {\label{tab:ids}List of wavelengths and identifications for the EUV lines. 
 } 
%
%\centerline
 {\tiny
\begin{tabular}{@{}lrlllll@{}}
 \hline\hline\noalign{\smallskip}  
 $i-j$   & $\lambda_{\rm exp.}$ &  $\lambda_{\rm obs.}$(\AA) &  ID & Diff. ID  \\ 
\noalign{\smallskip}\hline \noalign{\smallskip}
19--96  & 41.370 & 41.37(20) F79    &   F79    &         \\ %
3--47  & 46.300 & 46.30(20)  (bl) A85    &  N   &         \\ %
5--56  & 46.307 & 46.30(20) (bl) A85     &  N   &         \\ %
2--44  & 46.400 & 46.40(20) (bl) A85    &   N  &       \\ %
7--55  & 49.870 & 49.87(20) (bl)  F79    & F79         &    \\ %
8--53  & 50.260 & 50.26(20) F79 (bl)   & F79         &    \\ %
33--135  & 50.690 & 50.69(20) (bl) A85    &  N   &      \\ %
8--38  & 57.320 & 57.320(20) F79    &  F79   &        \\ %
33--119  & 57.710 & 57.71(20)  A85     &  N   &      \\ %
18--64  & 58.621 & 58.62(20) F79 (bl)    &  F79   &         \\ %
24--72  & 58.752 & 58.76(20) F79 (bl)   &   N  &   F79     \\ %
19--62  & 58.760 & 58.76(20) F79 (bl)   &  F79   &       \\ %
25--74  & 58.980 & 58.98(20) F79  (bl)   &   F79   &          \\ %
20--61  & 58.980 & 58.98(20) F79    &   F79  &         \\ %
26--73  & 59.259 & 59.26(20)  F79 (bl)  &  F79   &         \\ %
21--66  & 59.260 & 59.26(20) F79  (bl)  &   F79  &         \\ %
22--67  & 59.590 & 59.59(20)  F79   &  F79   &        \\ %
27--56  & 68.510 & 68.51(20) A85     &   N  &       \\ %
3--29  & 89.760 & 89.76(20) A85   &   N  &    Fi85 (87.30)    \\ %
3--15  & 204.668 & 204.668(10) N (bl)    &  D78   &         \\ %
6--18  & 254.536 & 254.536(10) N  (bl)    &  J84,B85(254.48)   &         \\ %
5--15  & 254.885 & 254.885(10) N (bl)      & D78(254.87)    &        \\ %
7--21  & 259.705 & 259.722(10) N  (bl)  &   N  &      \\ %
8--22  & 262.699 & 262.699(10) N  (bl)  &  N   &       \\ %
6--17  & 264.785 & (bl Fe XIV)    &  N   &    \\ %
12--24  & 266.417 & 266.417(10) N    &  J84,B85(266.43)   &         \\ %
7--20  & 269.420 & 269.420(10) N    &   J84  &  ?B85(269.61)       \\ % My EIS measurement.
6--16  & 269.886 & 269.886(10) N  (bl)    &  J84   &         \\ %
33--37  & 273.347 & 273.347(10)   N  (bl)     & N        &     \\ %
7--18  & 274.210 & (bl Fe XIV)    &  N   &      \\ %
9--21  & 275.550 & 275.550(10) N     &    J84,B85(275.60)  &         \\ %
8--20  & 279.245 & 279.240(10) N  (bl)  &  N   &       \\ %
14--26  & 280.160 & 280.160(10) N (sbl)  &   D08  &   J84(279.21)      \\ %
       &         &                       &       &     B85(279.1)      \\ %
10--22  & 280.160 & 280.160(10) N (sbl)    &  J84, B85, D08   &         \\ %
13--25  & 281.120 & 281.120(10) N     &   J84(281.09), B85    &       \\ %
8--19  & 283.942 & 283.945(10) N    &   N       & B85(284.01)    \\ %
  &                       &       &       &  J84(284.17)   \\ %
14--24  & 288.945 & 288.960(10) weak     & N            &   \\ %
3--11  & 295.981 & 295.98(20) D78     &  F85   &   J84(297.34)     \\ %
      &         &                       &       &     B85(297.46)      \\ %
10--18  & 304.971 & 304.971(10)  E N    &   B85(304.93)   &           \\ %
       &         &      (bl MnXIV, FeXV)                 &       &           \\ %
2--10  & 323.572 & 323.57(20)  D78   & J84, B85(323.65)   &         \\ %
4--13  & 340.122 & 340.12(20)  D78   &  F85   &  J84(337.23)      \\ %
3--10  & 340.391 & 340.40(20) D78    &  J84,B85(340.47)   &         \\ %
5--14  & 347.816 & 347.814(4) TN94  &  J84    &  B85(347.96)       \\ %
2--8  & 350.478 & 350.477(5) TN94    &   J84,B85(350.58)    &         \\ %
5--13  & 351.533 & 351.58(20) D78   &  F85, B85(351.69)    &         \\ %
3--9  & 358.247 & 358.247(6)  TN94   &   J84     &  ?B85       \\ %
2--7  & 367.288 & 367.287(8) TN94    &   F85,B85(367.37)    &  J84(365.62)      \\ %
4--12  & 373.430 & ?  373.41(20) F85   &   F85   &  B85(372.93)      \\ %
5--12  & 387.231 & 387.23(20)  D78   &  F85, B85(387.36)   &         \\ %
3--7  & 389.111 & 389.11(20)  (bl)  & F85, B85(389.25)   &  J84(387.23)      \\ %
2--6  & 409.705 & 409.705(6) TN94   & F85    &   J84(410.46)    \\ %
      &         &                       &       &    B85(409.91)      \\ %
3--4  & 1153.16 & 1153.16(20) F98    &  F85   &         \\ %
\noalign{\smallskip}
\hline                                
\end{tabular}}
\par
\smallskip
Notes: {See Table~\ref{tab:idsx}
for a description of the columns.  $\lambda_{\rm exp.}$ (\AA) is the experimental
wavelength obtained from the experimental energies. 
The estimated uncertainty in $\lambda_{\rm obs}$ in parentheses is in m\AA.
N in the $\lambda_{\rm obs}$ column indicates a new measurement 
based on the Hinode EIS observations presented here.
  (sbl) indicates the presence of a self-blend.
Some of the literature wavelengths~(\AA) are indicated in parentheses. 
References: D78: \cite{dere:78}; F79: \cite{fawcett_etal:79a};
 J84: \cite{jupen:84}; F85:\cite{feldman_etal:85};
B85: \cite{buchet_etal:85};  Fi85: \cite{finkenthal_etal:85};
 A85: \cite{acton_etal:85}; 
TN94:\cite{thomas_neupert:94}; 
 F98: \cite{feldman_etal:98}; D08:\cite{delzanna_bflare:08}.  
}
\end{table}